Author’s response to reviews

Title: Zinc finger protein ZBTB20 expression is increased in hepatocellular carcinoma and associated with poor prognosis

Authors:

Qing Wang (qing@hotmail.com)
Ye-xiong Tan (yexiong@yahoo.com.cn)
Yi-bin Ren (yibinren@126.com)
Li-wei Dong (donliwei@yahoo.com.cn)
Zhi-fang Xie (zhifangxie123@163.com)
Liang Tang (liangtang@yahoo.com)
Dan Cao (dancao@yahoo.com.cn)
Wei-ping Zhang (zhangwei1234@126.com)
He-ping Hu (hp-hu@medmail.com.cn)
Hong-yang Wang (hywangk@vip.sina.com)

Version: 3 Date: 1 June 2011

Author’s response to reviews: see over
Dear Editor:

It is our great pleasure to receive your positive reply!

As required, we have checked carefully the latest draft and made some change.(1) Eliminate the unnecessary elements of figures. (2) Renew the affiliation (3) include the tables in the manuscript in a tables section following the references. (4) add the authors list up to 30 authors names for each citation. 

Besides, at the “Acknowledgements and funds”, we added an fund (No. 30772478).

Note please: As we made some new changes as mentioned above, we reserved the “highlighting and tracking” in this revised manuscript. Please check it and then remove it.

Please tell us anything further needed. Thank you very much!

Best wishes!

Yours sincerely

Hongyang Wang